Uterotonic drugs to prevent postpartum haemorrhage: a network meta-analysis

Ioannis Gallos,¹* Helen Williams,¹ Malcolm Price,² Karen Pickering,² Abi Merriel,¹ Aurelio Tobias,¹ David Lissauer,¹ Harry Gee,¹ Özge Tunçalp,³ Gillian Gyte,^{4,5} Vidhya Moorthy,¹ Tracy Roberts,² Jonathan Deeks,² Justus Hofmeyr,⁶ Metin Gülmezoglu³ and Arri Coomarasamy¹

¹Tommy's National Centre for Miscarriage Research, Institute of Metabolism and Systems Research, College of Medical and Dental Sciences, University of Birmingham, Birmingham, UK

²Institute of Applied Health Research, College of Medical and Dental Sciences, University of Birmingham, Birmingham, UK

³Department of Reproductive Health and Research, World Health Organization, Geneva, Switzerland

⁴Department of Women's and Children's Health, Institute of Translational Medicine, University of Liverpool, Centre for Women's Health Research, Liverpool Women's NHS Foundation Trust, Liverpool, UK

⁵National Childbirth Trust, London, UK

⁶Effective Care Research Unit, University of the Witwatersrand/Fort Hare, Eastern Cape Department of Health, East London, South Africa

*Corresponding author i.d.gallos@bham.ac.uk

Declared competing interests of authors: Ioannis Gallos, Metin Gülmezoglu, Justus Hofmeyr and Arri Coomarasamy have been involved in one or more previous or ongoing trials related to the use of uterotonics for the prevention of postpartum haemorrhage that were considered for inclusion in this review. Ferring Pharmaceuticals (Saint-Prex, Switzerland) and Novartis Pharmaceuticals UK Ltd (Surrey, UK) have supplied carbetocin and oxytocin to these studies. Ioannis Gallos, Metin Gülmezoglu, Justus Hofmeyr and Arri Coomarasamy have not participated in decisions regarding inclusion of these trials in this review or any tasks related to them such as data extraction or quality assessment. Arri Coomarasamy is involved in a World Health Organization-sponsored randomised controlled trial of carbetocin versus oxytocin, supported by Merck for Mothers (Merck & Co., Inc., Kenilworth, NJ, USA). Metin Gülmezoglu was involved in a large multicentre trial included in the review as part of the central co-ordination unit. As part of the central co-ordination unit, he is also involved in an ongoing World Health Organization-sponsored randomised controlled trial of carbetocin versus oxytocin supported by Merck for Mothers. Abi Merriel is part-funded by Ammalife (a UK-registered charity 1120236) and the Birmingham Women's NHS Foundation Trust. Harry Gee and Arri Coomarasamy are trustees of Ammalife. Jonathan Deeks is a member of the Health Technology Assessment (HTA) Commissioning Board and the HTA Efficient Study and Designs Board.

Published February 2019 DOI: 10.3310/hta23090

Plain English summary

Uterotonic drugs to prevent postpartum haemorrhage

Health Technology Assessment 2019; Vol. 23: No. 9 DOI: 10.3310/hta23090

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

Postpartum haemorrhage (PPH) is the most common reason why mothers die in childbirth worldwide. Although most healthy women can cope well with blood loss after birth, some do not, and this can pose a serious risk to their health and even life. To reduce blood loss after birth, the routine administration of a drug to contract the uterus (uterotonic) has become standard practice across the world. This research seeks to identify which is the most effective and cost-effective drug.

Different drugs have been used for reducing the occurrence of PPH. They include oxytocin, misoprostol, ergometrine, carbetocin, and combinations of these drugs, each with different effectiveness and side effects. The study synthesised the available evidence to compare all of these drugs and combinations thereof. After putting the results of all available comparisons together in a network, a ranking among them was calculated, and provided robust effectiveness and side-effect profiles for each drug and their associated costs.

The study included 137 randomised trials, involving a total of 87,466 women. The results suggested that ergometrine plus oxytocin, carbetocin and misoprostol plus oxytocin are the most effective strategies for preventing PPH and are more effective than the currently recommended drug, oxytocin. Each of these three strategies had almost 100% probability of being ranked first, second or third most effective. Oxytocin was ranked fourth with an almost 0% probability of being ranked in the top three. Ergometrine plus oxytocin and misoprostol plus oxytocin were the worst drug combinations for side effects, with carbetocin having the most favourable side-effect profile. Carbetocin could prevent approximately one further event of PPH out of three in comparison with oxytocin. However, existing carbetocin studies were small and of poor quality. There is need for a large high-quality study comparing carbetocin with the current standard treatment of oxytocin for the prevention of PPH. The cost analyses of the alternative drug strategies remain inconclusive.

© Queen's Printer and Controller of HMSO 2019. This work was produced by Gallos et al. under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Health Technology Assessment

ISSN 1366-5278 (Print)

ISSN 2046-4924 (Online)

Impact factor: 4.513

Health Technology Assessment is indexed in MEDLINE, CINAHL, EMBASE, The Cochrane Library and the Clarivate Analytics Science Citation Index.

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full HTA archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hta. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Health Technology Assessment journal

Reports are published in *Health Technology Assessment* (HTA) if (1) they have resulted from work for the HTA programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Reviews in *Health Technology Assessment* are termed 'systematic' when the account of the search appraisal and synthesis methods (to minimise biases and random errors) would, in theory, permit the replication of the review by others.

HTA programme

The HTA programme, part of the National Institute for Health Research (NIHR), was set up in 1993. It produces high-quality research information on the effectiveness, costs and broader impact of health technologies for those who use, manage and provide care in the NHS. 'Health technologies' are broadly defined as all interventions used to promote health, prevent and treat disease, and improve rehabilitation and long-term care.

The journal is indexed in NHS Evidence via its abstracts included in MEDLINE and its Technology Assessment Reports inform National Institute for Health and Care Excellence (NICE) guidance. HTA research is also an important source of evidence for National Screening Committee (NSC) policy decisions.

For more information about the HTA programme please visit the website: http://www.nets.nihr.ac.uk/programmes/hta

This report

The research reported in this issue of the journal was funded by the HTA programme as project number 14/139/17. The contractual start date was in May 2015. The draft report began editorial review in November 2016 and was accepted for publication in May 2018. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HTA editors and publisher have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the draft document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health and Social Care.

© Queen's Printer and Controller of HMSO 2019. This work was produced by Gallos *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

NIHR Journals Library Editor-in-Chief

Professor Ken Stein Chair of HTA and EME Editorial Board and Professor of Public Health, University of Exeter Medical School, UK

NIHR Journals Library Editors

Professor Ken Stein Chair of HTA and EME Editorial Board and Professor of Public Health, University of Exeter Medical School, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals)

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Scientific Advisor, NETSCC, UK

Dr Catriona McDaid Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Professor of Wellbeing Research, University of Winchester, UK

Professor John Norrie Chair in Medical Statistics, University of Edinburgh, UK

Professor John Powell Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK

Professor James Raftery Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Professor Martin Underwood Warwick Clinical Trials Unit, Warwick Medical School, University of Warwick, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk